Author's response to reviews

Title: Acupuncture and chiropractic care for chronic pain in an integrated health plan: a mixed methods study

Authors:

Lynn L DeBar (lynn.debar@kpchr.org)
Charles Elder (charles.elder@kpchr.org)
Cheryl Ritenbaugh (ritenbau@email.arizona.edu)
Mikel Aickin (maickin@comcast.net)
Rick Deyo (deyor@ohsu.edu)
Richard Meenan (richard.meenan@kpchr.org)
John Dickerson (john.f.dickerson@kpchr.org)
Jennifer A. Webster (jennifer.a.webster@kpchr.org)
Bobbi Jo Yarborough (bobbijo.h.yarborough@kpchr.org)

Version: 3 Date: 23 November 2011

Author's response to reviews: see over
November 23, 2011

Dear Miss Puebla,

As requested we are resubmitting study protocol entitled, “Acupuncture and Chiropractic Care for Chronic Pain in an Integrated Health Plan: A Mixed Methods Study” with appropriate modifications. We have addressed all major and minor formatting changes requested by the journal and this uploaded version has been checked for any typographical errors.

The corresponding author is Lynn DeBar, PhD, MPH, Kaiser Permanente Center for Health Research, 3800 N. Interstate Avenue, Portland, Oregon 97227. (503-335-6796 telephone, 503-335-6311 FAX, lynn.debar@kpchr.org)

Sincerely,

Lynn DeBar, Ph.D., MPH